GOVERNMENT OF GUAM DEFERRED COMPENSATION PLAN

(Effective as of January 1, 2018)

GOVERNMENT OF GUAM DEFERRED COMPENSATION PLAN

TABLE OF CONTENTS

	<u>Page</u>
PREAMB	LE1
ARTICLE I	DEFINITIONSI-1
1.01	"Account"I-1
1.02	"Alternate Payee"
1.03	"Beneficiary"I-1
1.04	"Board of Trustees"I-1
1.05	"Code"
1.06	"Compensation"
1.07	"Deferral Election"I-1
1.08	"Director"
1.09	"Effective Date"I-1
1.10	"Elective Employer Contributions"
1.11	"Election Period"
1.12	"Employee"I-2
1.13	"Employer"I-2
1.14	"Employer Contributions"
1.15	"GCA"I-2
1.16	"Investment Options"I-2
1.17	"Investment Sponsors"I-2
1.18	"Normal Retirement Age"I-2
1.19	"Normal Retirement Date"
1.20	"Participant"I-3
1.21	"Plan"
1.22	"Plan Administrator"I-3
1.23	"Plan Year"I-3
1.24	"Qualified Custodian"I-3
1.25	"Qualified Domestic Relations Order"I-3
1.26	"Rollover Contributions"
1.27	"Roth Contributions"
1.28	"Service"I-3
1.29	"Spouse"I-3
1.30	"Trust"I-4
1.31	"Trustee"I-4
1.32	"Valuation Date"I-4
ARTICLE II	PARTICIPATION IN THE PLANII-1
2.01	Participation RequirementsII-1
2.02	Employees ExcludedII-1
2.03	Duration of ParticipationII-2

2.04 2.05	Reemployment	
ARTICLE III	•	
3.01 3.02	Elective Employer Contributions	III-1
ARTICLE IV	ALLOCATION TO ACCOUNTS	IV-1
4.01 4.02	Separate Accounts Limitation on Contributions	
ARTICLE V	VESTING OF ACCOUNTS	. V-1
ARTICLE VI	VALUATION OF ACCOUNTS	VI-1
6.01 6.02 6.03 6.04 6.05 6.06 6.07 6.08	Valuation Valuation of Segregated Accounts. Valuation Date Special Valuation Dates at Plan Administrator's Discretion. Method of Valuation Participant Accounts Statement of Account Valuation of Account When Payment Due	VI-1 VI-1 VI-1 VI-1 VI-2 VI-2
ARTICLE VI	I DISTRIBUTION OF BENEFITS	VII-1
7.01 7.02 7.03 7.04 7.05 7.06 7.07	Distribution Event	VII-1 VII-5 II-11 II-12 II-13
ARTICLE VI	II DESIGNATION OF BENEFICIARYV	III-1
8.01 8.02 ARTICLE IX	Designation Procedure V Lack of Designation V ADMINISTRATION	III-1
9.01 9.02 9.03 9.04 9.05 9.06	Plan Governance Plan Representatives Reliance Indemnification Expenses Claims for Benefits	IX-1 IX-2 IX-2 IX-3 IX-3
ARTICLE X	INVESTMENTS	. X-1
10.01 10.02 10.03	Funding Separate Investment Funds Loans to Participants	. X-1

ARTICLE XI	ASSIGNMENTS	XI-1
11.01	No Assignment	XI-1
11.02	Qualified Domestic Relations Order	XI-1
ARTICLE XII	AMENDMENT AND TERMINATION	XII-1
12.01	Amendment of Plan	XII-1
12.02	Termination of Plan	XII-1
12.03	Action Required Upon Plan Termination	XII-1
ARTICLE XII	I QUALIFIED MILITARY SERVICE REQUIREMENTS	XIII-1
13.01	General Employer Contributions	XIII-1
13.02	Employer Contributions	XIII-1
13.03	Interpretation	XIII-2
	HEART Act Provisions	
ARTICLE XI	MISCELLANEOUS	XIV-1
14.01	Participant's Rights	XIV-1
14.02	Construction	XIV-1
	Headings	
	Gender and Number	

GOVERNMENT OF GUAM DEFERRED COMPENSATION PLAN

PREAMBLE

The Government of Guam Deferred Compensation Plan is intended to advance the interests of Government of Guam by providing deferred compensation benefits to eligible Employees of the Employer, and has been established in accordance with Section 8302(a) of Title 4 of the Guam Code Annotated.

The Plan is intended to constitute an "eligible deferred compensation plan" meeting the requirements of Section 457(b) of the Internal Revenue Code of 1986, as amended. As a Code Section 457(b) "eligible deferred compensation plan", the Plan is exempt from the requirements of Code Section 409A pursuant to Code Section 409A(d)(2).

Under the terms of the Plan, originally effective May 1, 1999, the Employer reserves the right to amend the Plan at any time, and the Employer desires to amend the Plan in compliance with changes in law including the Job Creation and Workers Assistance Act of 2002, the Working Families Tax Relief Act of 2004, the Final Treasury Regulations under Code Sections 401(a)(9) and 457, the Pension Protection Act of 2006 ("PPA"), the Worker, Retiree, and Employer Recovery Act of 2008 ("WRERA"), and the Heroes Earnings Assistance and Relief Tax Act of 2008 ("HEART").

Further, the Employer desires to amend the Plan to comply with the requirements of Public Law No. 33-186, enacted by the Legislature of the Government of Guam on September 14, 2016, which establishes the "Defined Benefit 1.75 Retirement System" as codified in Article 5, Chapter 8, Title 4 of the GCA. Notwithstanding the preceding sentence however, the Plan shall not accept the transfer of amounts from accounts under the Government of Guam Defined Contribution Retirement System as described under GCA Section 8504, in order that the Plan may continue to meet the requirements of an eligible plan under Code Section 457 and Treasury Regulation Section 1.457-10.

Now therefore, based on the above, the Employer hereby amends and restates the Plan in its entirety as provided herein effective as of January 1, 2018, except as otherwise provided herein.

ARTICLE I

DEFINITIONS

As used in this Plan, the following terms shall have the meanings hereinafter set forth:

- 1.01 "Account" shall mean the aggregate of the accounts maintained for each Participant which represents his interest in the Plan and which shall include any account described in Section 4.01.
- 1.02 "Alternate Payee" shall mean any Spouse, former Spouse, child, or other dependent of a Participant recognized by a domestic relations order as having a right to receive all, or a portion of, a Participant's benefits under the Plan.
- 1.03 "Beneficiary" shall mean any person designated by a Participant to receive benefits upon the death of such Participant as described in Article VIII.
- 1.04 "Board of Trustees" shall mean the Board of Trustees of the Government of Guam Retirement Fund, as described GCA Section 8301(b).
- 1.05 "Code" shall mean the Internal Revenue Code of 1986, as amended from time to time, and mirror provisions of the Guam Territorial Income Tax Code.
- 1.06 "Compensation" shall mean all cash compensation for services to the Employer, including salary, wages, fees, commissions, bonuses, and overtime pay, that is includible in the Employee's gross income for the calendar year, plus amounts that would be cash compensation for services to the Employer includible in the Employee's gross income for the calendar year but for a compensation reduction election under Code Sections 125, 132(f), 401(k), 403(b), or 457(b), including an election to defer compensation under Article III. To the extent permitted by the Code, Treasury Regulations, or other similar guidance, Compensation shall include accrued bona fide sick, vacation or other leave pay paid within two and one-half months of a Participant's severance from Service so long as the employee would have been able to use the leave if employment had continued.
- 1.07 "<u>Deferral Election</u>" shall mean the election made by a Participant described in Section 3.01(a).
- 1.08 "<u>Director</u>" shall mean the Director of the Government of Guam Retirement Fund, as described in GCA Section 8301(d).
- 1.09 "Effective Date" shall mean January 1, 2018, the effective date of this amendment and restatement of the Plan, except as otherwise provided herein. The prior version of the Plan was effective as of January 1, 2002. The original effective date of the Plan was May 1, 1999.
- 1.10 "Elective Employer Contributions" shall mean the contributions to the Plan by the Employer as described in Section 3.01. However, the term shall mean only pre-tax Elective Employer Contributions (and not Roth Contributions) where the context so indicates.

- 1.11 "Election Period" shall mean the period within which a Deferral Election is made as described in Section 3.01(b).
 - 1.12 "Employee" shall mean any person employed by the Employer.
- 1.13 "Employer" shall mean the Government of Guam, and any entity in the Government of Guam described in GCA Section 8301(f), including the following government entities, organizations or departments that are organized and operated in Guam: each and every line department or agency of the Executive Branch; every autonomous and semi-autonomous agency or instrumentality; every public corporation; every educational institution, whether secondary or post-secondary; the Legislative Branch; the Judicial Branch; and the Public Defender Corporation, as those terms are defined in the GCA, and every public entity created by law, within Guam.
- 1.14 "Employer Contributions" shall mean the aggregate of contributions made to the Plan by the Employer as described in Article III.
- 1.15 "GCA" shall mean the Guam Code Annotated. Unless otherwise specifically provided, references to the GCA shall mean references to Title 4 of the GCA.
- 1.16 "Investment Options" shall mean the investment options made available by an Investment Sponsor and designated pursuant to the terms of the Plan as being available for the investment of Accounts under this Plan.
- 1.17 "Investment Sponsors" shall mean an insurance company, regulated investment company, or other entity providing Investment Options under the Plan.
- 1.18 "Normal Retirement Age" shall mean age 70½, unless the Participant has elected an alternate Normal Retirement Age and delivered such election to the Plan Administrator prior to beginning use of the Special Catch-Up Limitation as described in Section 4.02(c) of the Plan. Once a Participant has begun use of the Special Catch-Up Limitation, his Normal Retirement Age may not be changed.

In accordance with Treasury Regulation 1.457-4(c)(3)(v), for Participants eligible to receive benefits under the Employer's basic defined benefit pension plan or money purchase pension plan, a Participant's alternate Normal Retirement Age may not be earlier than the earliest date the Participant has the right to retire and receive immediate retirement benefits under such pension plan, without actuarial or similar reduction because of retirement before some later specified age, and the date selected may not be later than age $70\frac{1}{2}$.

If the Participant is not eligible to receive benefits under a basic defined benefit pension plan or money purchase pension plan, the Participant's alternate Normal Retirement Age may not be earlier than age 65 nor later than age 70½.

A special rule may be applied by the Plan Administrator to qualified police or firefighters under the Plan, if any, under which any qualified police or firefighter, as defined under Code Section 415(b)(2)(H)(ii)(I), who is participating in the Plan may choose a Normal Retirement Age that is not earlier than age 40 nor later than age 70½.

- 1.19 "Normal Retirement Date" shall mean the first day of the month coinciding with or next following the Participant's attainment of Normal Retirement Age.
- 1.20 "Participant" shall mean any Employee or former Employee who has entered the Plan in accordance with Article II, and whose Account hereunder has not subsequently been distributed.
- 1.21 "Plan" shall mean Government of Guam Deferred Compensation Plan, as described in GCA Section 8301(a), and as set forth herein and as amended from time to time.
- 1.22 "Plan Administrator" shall mean the person, persons or entity appointed by the Employer to administer the Plan, which shall be the Director pursuant to Section 9.01.
 - 1.23 "Plan Year" shall mean each 12-month period ending on December 31.
- 1.24 "Qualified Custodian" shall mean a bank, as described in Code Section 408(n), a person other than a bank satisfying the requirements of Treasure Regulation Section 1.408-2(e) relating to the use of non-bank trustees, or the Board of Trustees in its capacity as Trustee, as described in GCA Section 8301(g).
- 1.25 "Qualified Domestic Relations Order" shall have the meaning set forth in Code Section 414(p).
- 1.26 <u>"Rollover Contributions"</u> shall mean such contributions to the Plan by the Employee as described in Section 3.02.
- 1.27 "Roth Contributions" shall mean Elective Employer Contributions that are designated by the Employee as after-tax Roth Contributions as described in Section 3.01(a).
- 1.28 "Service" shall mean an Employee's total period of employment with the Employer.
- 1.29 "Spouse" shall mean, the person to whom the Participant is married if the individuals are in a marriage that is recognized as a legal marriage under the Code. Under this definition, and in accordance with IRS Notice 2014-19 and Revenue Ruling 2013-17:
 - (a) Effective September 16, 2013, a legal marriage shall include a marriage of individuals of the same sex if the marriage is validly entered into in a state whose laws authorize same-sex marriage, even if they are domiciled in a state that does not recognize the validity of such marriages.
 - (b) Prior to September 16, 2013, and effective as of June 26, 2013, a legal marriage shall include a marriage of individuals of the same sex if the marriage is validly entered into in a state whose laws authorize same-sex marriage, but only if the participant is domiciled in a state that recognizes same-sex marriages.

For this purpose, the term "state" shall mean any state of the United States, the District of Columbia, any territory or possession of the United States, or any foreign jurisdiction having the legal authority to sanction marriages.

- 1.30 "Trust" shall mean the legal entity established pursuant to a written agreement that constitutes a valid trust under applicable state law and held by the Trustee. Trust shall also mean any annuity contracts or custodial accounts described in Section 10.01.
- 1.31 "Trustee" shall mean the Board of Trustees and any successor duly appointed pursuant to a trust agreement as described in Section 10.01.
- 1.32 "<u>Valuation Date</u>" shall mean the last day of each Plan Year and such other dates as may be designated as provided in Article VI for valuation of Participants' Accounts.

ARTICLE II

PARTICIPATION IN THE PLAN

2.01 Participation Requirements

(a) Eligibility Requirements

GCA Section 8301(e) provides that any person who is employed by the Employer and who is a member of either the Government of Guam Retirement Fund established in Article 1, Chapter 8 of the GCA or the Government of Guam Defined Contribution Retirement System established in Article 2, Chapter 8 of the GCA, with the exception of any person who participates under a Code Section 403(b) plan sponsored by the Employer, shall be eligible to participate in the Plan.

Accordingly, subject to any applicable statutory or regulatory exceptions, any full-time Employee of the Employer, any permanent, seasonal, temporary or similar part-time Employee of the Employer, and any elected or appointed official of the Employer, shall be eligible to participate in the Plan. However, any Employee who is included in a unit of employees covered by a collective bargaining agreement that does not specifically provide for participation in the Plan shall be excluded from participation in the Plan. Independent contractors shall also be excluded from participation in the Plan.

(b) <u>Date of Participation</u>

(1) Effective Date

Each Employee who is a Participant in the Plan immediately prior to the Effective Date of this Plan amendment and restatement shall continue to participate in the Plan as of the Effective Date.

(2) After Effective Date

Each Employee not meeting the above Section 2.01(b)(1) shall become a Participant in the Plan as of the first day of the calendar month coinciding with or first following the date on which the Employee meets the eligibility requirements described in Section 2.01(a).

Notwithstanding qualification as a Participant under this Section 2.01, no Participant shall have any rights or interests under the Plan until becoming an active Participant by having also made a contribution to the Plan described in Article III. Active participation in the Plan and all contributions shall be voluntary, in accordance with GCA Section 8302(b).

2.02 Employees Excluded

If an Employee participating in the Plan thereafter fails to meet the eligibility requirements of Section 2.01, Employer Contributions on behalf of the

Employee shall cease effective as of the effective date of such failure. While an excluded Participant, the Employee's Account shall continue to share in the allocation of gains and losses under the Plan and shall be distributed pursuant to the applicable provisions of the Plan.

If an Employee fails to meet the eligibility requirements of Section 2.01 and subsequently meets such requirements, the Employee shall be eligible to participate in the Plan as of the date on which he meets the eligibility requirements described in Section 2.01.

2.03 <u>Duration of Participation</u>

An Employee who becomes a Participant in the Plan shall remain a Participant (even after termination from Service) for so long as he maintains an Account under which he is entitled to receive benefits under the Plan.

2.04 Reemployment

A former Employee may become a Participant in the Plan after his reemployment with the Employer in accordance with the participation requirements set forth in Section 2.01.

2.05 <u>Termination of Participation</u>

A Participant shall cease to be a Participant as of the date the Participant receives a total distribution of the vested interest in his Account.

ARTICLE III

CONTRIBUTIONS

3.01 Elective Employer Contributions

(a) <u>Deferral Election</u>

In accordance with GCA Section 8302(b), participation in the Plan shall be voluntary. Each Participant may make an election ("Deferral Election") to have the Employer employing him reduce his Compensation for each payroll period by an amount equal to a whole percentage or a specified dollar amount, subject to the limitations on contributions as described in Section 4.02. For each payroll period, the Employer shall contribute to the Plan on behalf of each Participant an amount equal to the amount of Compensation which the Participant elects to reduce pursuant to his Deferral Election for the payroll period ("Elective Employer Contributions"). A Deferral Election shall not apply to any Compensation currently available on or before the later of the date on which this Section 3.01 is adopted by the Employer as part of the Plan or the date on which the provisions of this Section 3.01 first becomes effective. Elective Employer Contributions made on a Participant's behalf shall be allocated to the Elective Contribution Account of the Participant described in Section 4.01.

As soon as administratively practicable after the adoption date of this amendment, a Participant shall also be allowed to designate on the Deferral Election the type of Elective Employer Contributions (either after-tax Roth Contributions, pre-tax Elective Employer Contributions, or a combination of both as such contributions are described in this Section 3.01) to be withheld.

(1) <u>Designated Roth Contributions</u>

As soon as administratively practicable after the adoption date of this amendment, a Participant shall be allowed to irrevocably designate on the Participant's Deferral Election that a portion of the Participant's Elective Employer Contributions shall be Roth Contributions ("Roth Contributions"). Such designated contributions made on behalf of a Participant shall constitute designated Roth contributions within the meaning of Code Section 402A to the extent the following requirements are satisfied.

(A) After-Tax Treatment

The designated Roth Contributions are treated by the Employer as includible in the Participant's income (e.g., treated as wages subject to applicable withholding requirements) at the time the Participant would have received such contributions as cash but for the Deferral Election.

(B) Separate Account

The Roth Contributions are credited to and maintained under the Participant's Roth Contribution Account described in Section 4.01(c) as a separate account from the Participant's Elective Contribution Account.

The Plan Administrator may at its discretion determine on a uniform and nondiscriminatory basis whether the Participant's designation of Roth Contributions may apply to all or a limited portion of the Elective Employer Contributions. To the extent that the Participant does not designate his Elective Employer Contributions as Roth Contributions, such Elective Employer Contributions shall constitute pre-tax Elective Employer Contributions under the Plan.

Except as otherwise provided hereunder, the Participant's Roth Contributions shall constitute Elective Employer Contributions and shall be subject to all Plan requirements applicable to Elective Employer Contributions.

(2) <u>Defined Benefit 1.75 Retirement System Participants</u>

Notwithstanding the foregoing portion of this Section 3.01(a), mandatory deferrals and contributions to the Plan, as described in this Section 3.01(a)(2), shall be required for any Participant who is a member of the Defined Benefit 1.75 Retirement System ("Defined Benefit 1.75 Plan"), as defined in GCA Section 8501(j). In accordance with GCA Section 8508, for each Plan Year, the Employer of a Participant who is also a member of the Defined Benefit 1.75 Plan shall reduce his Compensation by an amount equal to 1%, and shall contribute an equal amount to the Plan on behalf of the Participant. Reduction and contribution rates may be adjusted by the Employer (and may exceed 1%), as necessary, to comply with the contribution requirement of GCA Section 8508. An election to participate in the Defined Benefit 1.75 Plan shall be deemed to constitute a Deferral Election for such deferral and contribution as described in this Section 3.01, and constitute an agreement providing for such deferral, as described in Code Section 457(b)(4).

Plan contributions required under this Section 3.01(a)(2) shall be administered as pre-tax Elective Employer Contributions, and shall constitute pre-tax "pick-up" employer contributions comparable to those described in Code Section 414(h). Although such contributions shall be designated as employee contributions, such contributions shall be paid to the Plan by the Employer. A Participant shall not have a cash or deferred election right with respect to such contribution amounts as of the date of the "pick-up". Specifically, a Participant shall not be permitted to opt-out of the "pick-up" arrangement, or opt to receive such contribution amounts directly.

(b) Deferral Election Procedure

Each Participant shall make a Deferral Election by submitting the prescribed election form to the Employer during the applicable election period described hereunder ("Election Period"). In accordance with GCA Section 8302(b), the amount to be deferred must be as provided in a written agreement timely executed by and between the Participant and the Employer. Such agreement shall be in a form as prescribed by the Director, approved by the Board of Trustees, and in compliance with the requirements of Code Section 457.

With respect to the first Plan Year, the Election Period shall be the period as determined by the Plan Administrator, and the Deferral Election submitted during such Election Period shall be effective as soon as practicable on or after the date on which the Deferral Election is submitted, but no earlier than the first payment of Compensation by the Employer during the first month following receipt by the Employer of the Deferral Election, in accordance with Code Section 457(b)(4) and Treasury Regulation 1.457-4(b).

For any Employee who becomes a Participant during the Plan Year, the applicable Election Period shall be a period, as determined by the Plan Administrator, immediately preceding the date such Employee commences participation in the Plan. A Deferral Election submitted during such Election Period shall be effective as soon as practicable on or after the date of the Employee's commencement of participation in the Plan, but no earlier than the first payment of Compensation by the Employer during the first month following receipt by the Employer of the Deferral Election, in accordance with Code Section 457(b)(4) and Treasury Regulation 1.457-4(b). However, a new Employee may defer compensation payable in the calendar month during which the Participant first becomes an Employee if a Deferral Election providing for the deferral is entered into by the Participant and approved by the Plan Administrator on or before the first day on which the participant performs services for the Employer, in accordance with Treasury Regulation 1.457-4(b). Once made, a Deferral Election shall remain in effect from Plan Year to Plan Year unless otherwise modified or terminated as provided herein.

Any prior Employee who was a Participant in the Plan and is rehired by Employer may resume participation in the Plan by entering into a Participation Agreement, however, any deferred commencement date elected by such Employee with respect to those prior Plan assets shall be null and void.

For purposes of this Article III, and in accordance with procedures established by the Plan Administrator, the prescribed election form for a Deferral Election (or the modification or termination of a Deferral Election) may be completed and delivered in electronic or written form to the Plan Administrator or the Plan's recordkeeper or other appropriate vendor on behalf of the Plan.

The Plan Administrator retains the right to establish minimum deferral amounts per payroll period, and to change such minimums from time to time and to limit the number or timing of enrollments into the Plan. No adjustment in future deferrals shall be made if a periodic deferral is missed or is less than the amount elected. The Plan

Administrator shall have the right to modify or disallow the periodic deferral of Compensation elected by the Participant:

- (1) in excess of the limitations stated in Article IV;
- (2) in excess of the Participant's net Compensation for any payroll period;
- (3) upon any change in the length of the payroll period utilized by the Employer. In such case the periodic deferral may be adjusted so that approximately the same percentage of pay shall be deferred on an annual basis;
- (4) in order to round periodic deferrals to the nearest whole dollar amount;
- (5) to reduce the future deferrals in the event that the amount actually deferred for any payroll period exceeds, for any reason whatsoever, the amount elected by the Participant. In the alternative, such amount of excess deferral may be refunded to the Participant; or
- (6) if the deferral elected for any payroll period is less than the minimum amount specified by the Plan Administrator.

(c) <u>Modification or Termination of Deferral Election</u>

(1) Modification

A Participant may modify a Deferral Election in effect for him at the time by submitting a new election form to the Plan Administrator, in which case the modified Deferral Election shall be effective as soon as administratively feasible following the date of such submission as determined by the Plan Administrator, but no earlier than the first payment of Compensation by the Employer during the first month following receipt by the Employer of the new election.

For the Plan Year in which Roth Contributions are initially implemented, Participants shall be allowed to modify their Deferral Elections at the time and manner determined by the Plan Administrator at its complete discretion.

(2) Termination

A Participant may terminate a Deferral Election at any time by submitting a new election form to the Plan Administrator for the discontinuation of Elective Employer Contributions on his behalf, in which case the termination of his Deferral Election shall be effective as soon as administratively feasible following the date of such submission as determined by the Plan Administrator,

but no earlier than the first payment of Compensation by the Employer during the first month following receipt by the Employer of the notice.

(d) Payment of Elective Employer Contributions

Elective Employer Contributions made on behalf of a Participant for any Plan Year shall be paid to the Trust by the Employer as described in Section 10.01 as soon as reasonably practicable following the date such contributions would have been paid to the Participant if the Participant made no Deferral Election, in accordance with Code Section 457(g).

3.02 Rollover Contributions and Direct Rollovers

The Plan Administrator may accept a rollover contribution or a direct rollover to this Plan of an Eligible Rollover Distribution (as defined in Section 7.02(b)(2)(A)) on behalf of an Employee, including an Employee who has not yet satisfied the eligibility requirements under Section 2.01 from an Eligible Retirement Plan (as defined in Section 7.02(b)(2)(B)), in accordance with Treasury Regulation 1.457-10(e). If the Plan accepts a rollover contribution or a direct rollover from more than one Eligible Retirement Plan, a separate Rollover Contribution Account shall be established and maintained to separately account for the rollover contribution or direct rollover from each such Eligible Retirement Plan. The Plan Administrator may, but need not, authorize any rollover contribution or direct rollover, and the Plan Administrator may prohibit all such rollovers at its complete discretion. Such rollover contribution or direct rollover shall comply with the requirements of Code Section 402, including a direct rollover of an Eligible Rollover Distribution pursuant to Code Section 401(a)(31), and a rollover from a conduit individual retirement account pursuant to Code Section 408(d)(3)(A). A rollover contribution or direct rollover shall be allocated to the Participant's Rollover Contribution Account as of the date such contribution or rollover is received.

(a) Information Required

As determined by the Plan Administrator, a rollover contribution or direct rollover shall be accompanied by written representations identifying the transferor plan, stating the name of the Employee, stating the current value of the assets to be transferred, stating that the Employee maintains a 100% fully vested interest in such assets under the transferor plan, and providing such other information as the Plan Administrator may require. The Plan Administrator, in its sole discretion, may require an opinion of legal counsel that the amounts to be transferred meet the requirements of this Section 3.02.

(b) Eligibility Requirements

In the event of a rollover contribution or direct rollover on behalf of an Employee who has not satisfied the eligibility requirements under Section 2.01, the Employee shall be treated as a Participant under the Plan, except that the Employee shall not be eligible to receive any Employer Contributions until the Employee satisfies the applicable eligibility requirements and allocation requirements under the Plan.

(c) <u>Investments</u>

The Plan Administrator may allow or restrict, in its sole discretion, that the amounts transferred be in cash, in-kind, or a combination thereof. The amounts transferred shall share in the earnings and losses of the Trust and shall be subject to the investment requirements as provided under the Plan. Therefore, in the event that the Plan Administrator accepts a rollover contribution or direct rollover in-kind, the Plan Administrator may direct the Trustee to invest such amount in a segregated Account for the sole benefit of the Employee pursuant to Article X, without regard to whether the Employee is allowed to otherwise segregate or self-direct the investment of any other of his Accounts.

(d) Rollover Characteristics

In the event of a rollover contribution or a direct rollover of an amount to this Plan from another plan, the protected benefits available under the transferor plan shall not be retained under this Plan, and the distribution options as permitted under this Plan shall apply with respect to the distribution of the transferred amount.

(e) <u>Transfer to the Plan due to Merger or Consolidation</u>

At the discretion of the Plan Administrator and subject to the requirements of this Section 3.02(e), the Plan may accept a transfer of assets from another eligible governmental plan within the meaning of Code Section 457(b) and Treasury Regulation Section 1.457-2(f).

Such transfer may only be allowed if the transferring plan is located within Guam, all of the assets held by the transferor plan are transferred, the transferor plan provides for transfers, a participant or beneficiary whose amounts deferred are being transferred will have an amount deferred immediately after the transfer at least equal to the amount deferred with respect to that participant or beneficiary immediately before the transfer, and the participants or beneficiaries whose deferred amounts are being transferred are not eligible for additional annual deferrals in the Plan unless they are performing services for the Employer.

The transferred amounts shall be subject to distribution restrictions of Treasury Regulation Section 1.457-6 under the Plan in the same manner as if the transferred amount had originally been deferred under the Plan if the participant is performing services for the Employer. Such transfer shall be carried out in accordance with the requirements of Treasury Regulation Section 1.457-10(b).

ARTICLE IV

ALLOCATION TO ACCOUNTS

4.01 Separate Accounts

(a) In General

The Plan Administrator shall establish and maintain separate Accounts for each Participant in accordance with GCA Section 8305(f), including an Elective Contribution Account and a Rollover Contribution Account as described in this Section 4.01.

As soon as administratively practicable after the adoption date of this amendment, the Plan Administrator shall also establish and maintain separate Roth Contribution Accounts for each Participant as described in this Section 4.01.

(b) Elective Contribution Account

The Plan Administrator shall establish and maintain an Elective Contribution Account for each Participant, to which shall be allocated pre-tax Elective Employer Contributions made on behalf of the Participant and any adjustment for net income or loss under the Trust attributable to such Account. Amounts under this Section 4.01(b) shall be allocated, as of the last day of the Plan Year (or more frequently as determined by the Plan Administrator), among the Elective Contribution Accounts of Participants in accordance with their Deferral Elections described in Section 3.01.

(c) Roth Contribution Account

As soon as administratively practicable after the adoption date of this amendment, the Plan Administrator shall establish and maintain a Roth Contribution Account for each Participant who designates a Roth Contribution to the Plan in accordance with Section 3.01(a), to which shall be allocated the Roth Contribution and any adjustments for net income or loss under the Trust attributable to such Account. The Roth Contribution Account shall be subject to separate accounting, under which contributions and withdrawals of Roth Contributions on behalf of a Participant shall be credited and debited to the Roth Contribution Account of such Participant. The Roth Contribution Account shall include a record of the Participant's "investment in the contract" (i.e., Roth Contributions that have not been distributed) under such Account. Gains, losses, and other credits or charges shall be separately allocated on a reasonable and consistent basis to the Roth Contribution Account and other Accounts. However, any Forfeitures under this Plan shall not be allocated to any Roth Contribution Account. The separate accounting requirements described in this Section 4.01(c) shall be applicable at the time any Roth Contribution is made and credited to a Participant's Roth Contribution Account and shall continue until such Roth Contribution Account is completely distributed.

- (1) <u>In-Plan Roth Conversion Election</u>. A Participant may elect to transfer ("In-Plan Roth Conversion") all or a portion of his vested interest in his Elective Contribution Account and Rollover Contribution Account to the Participant's Roth Contribution Account in accordance with the provisions of the Plan. In the event of an In-Plan Roth Conversion, the Participant shall continue to be eligible to participate in the Plan on the same basis as any other Participant. Except as otherwise provided under this Section 4.01(c)(1), and subject to the distribution requirements of Section 7.02(f)(3), In-Plan Roth Conversion amounts shall otherwise be treated as Roth Contributions in accordance with the terms of the Plan.
- (2) After-Tax Treatment. The taxable amount of an In-Plan Roth Conversion is treated by the Employer as includible in the Participant's income for the taxable year in which the transfer occurs. This amount is equal to the fair market value of the In-Plan Roth Conversion reduced by any basis of the Participant in the transferred amount. If a loan to a Participant, as described in Section 10.03, is transferred as a result of an In-Plan Roth Conversion, the taxable amount shall be the outstanding balance of the loan.
- (3) In-Plan Roth Transfer Credited to Roth Contribution Account. The election of an In-Plan Roth Conversion does not constitute a Plan distribution, or an election to receive an actual payment from the Plan to the Participant, and no such payment shall be made pursuant to the election. Rather, the amount of the In-Plan Roth Conversion shall be converted from the given existing Plan account and credited to the Participant's Roth Contribution Account. Accordingly, the Plan shall not withhold or distribute any amounts for income tax withholding, unless a distribution of other amounts is permitted pursuant to the terms of the Plan. The Plan shall transfer investments to the Participant's Roth Contribution Account in accordance with the Plan terms and procedures governing Plan investments. A Participant loan that is transferred to a Participant's Roth Contribution Account with no alteration to the repayment schedule shall not be treated as a new loan.
- (4) <u>Beneficiaries and Alternate Payees</u>. An In-Plan Roth Conversion may be elected by a Beneficiary only if he is a surviving Spouse, and by an Alternate Payee only if he is a Spouse or former Spouse.
- (5) <u>In-Plan Roth Transfer Account</u>. An "In-Plan Roth Conversion Account" is a subaccount of a Participant's Roth Contribution Account that the Plan Administrator establishes for the purpose of separately accounting for a Participant's transfers attributable to the Participant's In-Plan Roth Conversions. The Plan Administrator shall have authority to establish such a subaccount, and to the extent necessary, may establish subaccounts based on the source of the In-Plan Roth Conversion. The Plan Administrator will administer any In-Plan Roth Conversion Account in accordance with guidance provided by the Internal Revenue Service and pursuant to the terms of the Plan.

(d) Rollover Contribution Account

The Plan Administrator shall establish and maintain a Rollover Contribution Account for each Participant who makes a rollover contribution or direct rollover to the Plan in accordance with Section 3.02, to which shall be allocated such rollover contribution or direct rollover and any adjustments for net income or loss under the Trust attributable to such Account.

4.02 Limitation on Contributions

For purposes of this Section 4.02 only, in accordance with Treasury Regulation Section 1.457-2(g), Compensation shall mean, with respect to a taxable year, a Participant's compensation as defined in Code Section 415(c)(3), for services performed for the Employer. The amount of Compensation shall be determined without regard to any community property laws.

Compensation shall include a Participant's actual wages as reported in Box 1 of Internal Revenue Service Form W-2 for services to the Employer with respect to a taxable year, subject to a maximum of \$265,000 (or such higher maximum as may apply under Code Section 401(a)(17)) and increased (up to the dollar maximum) by any compensation reduction election under Code Sections 125, 132(f), 401(k), 403(b), or 457(b) (including an election to defer Compensation under Article III).

(a) Primary Limitation

The maximum amount ("Primary Limitation") that normally may be contributed to the Plan pursuant to contributions as described in Article III on behalf of a Participant shall be the lesser of: (1) the applicable dollar amount, as set forth in Code Section 457(e)(15) (e.g., \$18,000 for 2017); or (2) 100% of the Participant's Compensation for the Plan Year. Such limit will be adjusted by the Secretary of the Treasury for cost-of-living increases under Code Section 457(e)(15)(B). The Primary Limitation does not include any rollover amounts received by the Plan under Treasury Regulation Section 1.457-10(e).

(b) Age 50 Catch-Up Limitation

In addition to those contributions permissible under the Primary Limitation, additional amounts may be contributed to the Plan on behalf of a Participant who will attain age 50 or greater by the end of the Plan Year ("Age 50 Catch-Up Contributions") up to the limit described in Code Section 414(v)(2) ("Age 50 Catch-Up Limitation"), which is \$6,000 for 2017. Such limit will be adjusted by the Secretary of the Treasury for cost-of-living increases under Code Section 414(v)(2)(C).

(c) Special Catch-Up Limitation

For each of the last three Plan Years ending before a Participant's attainment of his Normal Retirement Age, if the amount determined under this Section 4.02(c) exceeds the total of the Primary Limitation and the Age 50 Catch-Up Limitation

described in Sections 4.02(a) and 4.02(b), the maximum amount that may be contributed to the Plan on behalf of the Participant shall be the lesser of: (1) twice the applicable dollar amount in effect under Code Section 457(b)(2)(A) for the Plan Year; or (2) the sum of (i) the Primary Limitation determined under Section 4.02(a) above for the Plan Year and (ii) that portion of the Primary Limitation determined under Section 4.02(a) above not utilized by the Participant in prior Plan Years in which the Participant was eligible to participate in the Plan ("Special Catch-Up Limitation").

The Special Catch-Up Limitation shall apply to a single three-year period with respect to any Participant. If a Participant utilizes the Special Catch-Up Limitation and postpones his retirement beyond his Normal Retirement Date or is reemployed following his retirement on or after his Normal Retirement Date, then the Special Catch-Up Limitation shall not apply again to such Participant.

(d) <u>Coordination of 50 Catch-Up Limitation and Special Catch-Up Limitation</u>

In accordance with Treasury Regulation 1.457-4(c)(2), a Participant who is eligible for Age 50 Catch-Up Contributions for a Plan Year, and for whom that Plan Year is also one of the Participant's last three Plan Years ending before the Participant attains Normal Retirement Age, the maximum amount that may be contributed to the Plan on behalf of a Participant shall be the greater of:

- (1) The total of the Primary Limitation described in Section 4.02(a) and the Age 50 Catch-Up Limitation described in Section 4.02(b); and
 - (2) The Special Catch-Up Limitation described in Section 4.02(c).

(e) Coordination With Other 457(b) Plans

If a Participant participates in more than one Code Section 457(b) plan, the maximum amount that may be contributed to all such plans for a Plan Year shall not exceed the applicable limitation described in Section 4.02(a) (subject to modification by the Age 50 Catch-Up Limitation and the Special Catch-Up Limitation described in Sections 4.02(b) and 4.02(c)).

(f) Excess Contributions

To the extent that any amount contributed for any Plan Year on behalf of a Participant exceeds the applicable maximum limitation described in this Section 4.02, such excess, as adjusted for allocable net income, shall be distributed to the Participant as soon as administratively practicable after the Plan determines that the amount is an excess deferral in accordance with Treasury Regulation Section 1.457-4(e)(2). Excess amounts shall be distributed in the following order of priority: (1) pre-tax Elective Employer Contributions; and (2) Roth Contributions.

ARTICLE V

VESTING OF ACCOUNTS

A Participant's interest in the balance of his Elective Contribution Account, Roth Contribution Account, and Rollover Contribution Account shall be fully and immediately vested and nonforfeitable at all times.

ARTICLE VI

VALUATION OF ACCOUNTS

6.01 <u>Valuation</u>

As of each Valuation Date, prior to any allocation of Employer Contributions to be made as of such date, the net income or loss of the Plan's general assets since the preceding Valuation Date, including net appreciation or depreciation and any expenses paid under the Plan, shall be allocated to each Account in the ratio that the value, as of the next preceding Valuation Date, of each such Account invested in the general assets bears to the value, as of the next preceding Valuation Date, of all Accounts invested in the general assets. If one or more separate investment funds have been established for purposes of the Plan, the net income or loss of each investment fund shall be similarly but separately allocated to each Account invested in such investment fund in proportion to the value of each Account invested in such funds as of the preceding Valuation Date. The Plan Administrator shall adopt equitable procedures to establish a proportionate crediting of Plan income or loss to those portions of Participants' Accounts in the case of contributions, transfers, or withdrawals that have occurred in the interim period since the next preceding Valuation Date.

6.02 <u>Valuation of Segregated Accounts</u>

The portion of any Participant's Account invested on a segregated basis as provided in Article X shall be valued separately on each Valuation Date and the net income or loss allocated to such Account shall be based on the assets, including income, gain, loss, allocated expenses, and other change in value of the assets constituting such portion of the Account.

6.03 Valuation Date

The General Account, any separate investment funds, and any segregated Account shall be valued as of the last day of each Plan Year.

6.04 Special Valuation Dates at Plan Administrator's Discretion

The Plan Administrator may determine the fair market value of an Account as of any date other than the last day of a Plan Year in accordance with the provisions of Section 6.02.

6.05 Method of Valuation

All determinations made by the Plan Administrator with respect to the value of an Account shall be made in accordance with generally accepted principles of accounting, and such determinations when so made by the Plan Administrator shall be conclusive and binding upon all persons having an interest under the Plan.

6.06 Participant Accounts

The Plan Administrator shall open and maintain separate Accounts for each Participant as described in Article IV, in accordance with GCA Section 8305(f). Each Participant's Account shall reflect the amounts allocated thereto and distributed therefrom and such other information as affects the value of such Account pursuant to the Plan. The records of the Plan shall be maintained by the Plan Administrator and shall accurately disclose the value of the Participant's Account.

6.07 Statement of Account

As soon as practicable after the end of each Plan Year, and the first day of the fourth, seventh, and tenth month of each Plan Year, the Plan Administrator shall furnish to each Participant a statement of his Account, determined as of the last day of such previous quarter of the Plan Year, in accordance with GCA Section 8309. Upon the discovery of any error or miscalculation in an Account, the Plan Administrator shall correct it, to the extent correction is practically feasible. Statements to Participants are for reporting purposes only, and no allocation, valuation, or statement shall vest any right or title to any part of the assets of the Employer, nor require any segregation of the assets of the Employer, except as is specifically provided in the Plan.

6.08 Valuation of Account When Payment Due

The amount of any payment in the case of a distribution from a Participant's Account pursuant to Article VII shall be based on the value of the Participant's Account as of the date of the applicable distribution event, in accordance with Section 8308 of the GCA, subject to adjustments pursuant to administrative rules and regulations under the Plan that are adopted by the Plan Administrator.

ARTICLE VII

DISTRIBUTION OF BENEFITS

7.01 Distribution Event

In the event of a Participant's severance from Service, the Participant (or, in the event of the Participant's death, his Beneficiary) shall be entitled to receive a benefit equal to the vested interest in his Account, the value of which shall be determined pursuant to Article VI. Distribution of the benefit shall commence at the time and form as described in this Article VII following the Participant's severance from Service, and shall be in accordance with the requirements of Treasury Regulation Section 1.457-6.

7.02 Manner of Distribution

(a) Time and Form of Distribution

Subject to applicable law and other provisions of this Plan, distributions may be made in accordance with one of the following payment options:

- (1) A single lump-sum payment;
- (2) Installment payments for a period of years (payable on a monthly, quarterly, semi-annual, or annual basis) which extends no longer than the life expectancy of the payee as permitted under Code Section 401(a)(9) (using the "declining balance method", under which each installment payment is determined by multiplying the remaining balance by a fraction, where the numerator is one and the denominator is the remaining payments in the payment period);
- (3) A partial lump-sum payment of a designated amount, with the balance payable in installment payments for a period of years, as described in Section 7.02(a)(2);
- (4) Annuity payments (payable on a monthly, quarterly or annual basis) for the lifetime of the payee or for the lifetimes of the payee and Beneficiary in compliance with Code Section 401(a)(9);
- (5) Such other forms of installment payments as may be approved by the Employer consistent with the requirements of Code Section 401(a)(9); or
- (6) A Participant who is an eligible retired public safety officer, as defined under Code Section 402(1)(4)(B), may elect to have distributions made directly to an insurer to pay qualified health insurance premiums for coverage for the eligible retired public safety officer, his Spouse and dependents, by an accident of health insurance plan or qualified long-term care insurance contract as defined in Code Section 7702B(b). Any elections and distributions under this Section 7.02(a)(6) shall be made in a manner consistent with the requirements and

limits contained in Code Section 402(1) and any applicable guidance issued thereunder.

A Participant shall make any distribution election in such manner and form as is approved by the Plan Administrator. The Employer or Plan Administrator may also adopt practices and procedures applicable to existing and new distribution elections. A payee's election of a payment option must be made at least 30 days prior to the date that the payment of benefits is to commence.

In the absence of an effective election by the Participant as to the commencement or form of benefits, distributions shall be made in accordance with the applicable requirements of Code Sections 401(a)(9) and 457(d), and final Treasury Regulations thereunder. In the absence of an effective election by the Beneficiary or Alternate Payee as to the commencement or form of benefits, distributions shall be made in a lump sum.

Notwithstanding any other provision of this Article VII, no distribution option may be selected by a payee under this Article VII unless it satisfies the applicable requirements of Code Sections 401(a)(9) and 457(d), and final Treasury Regulations thereunder. The terms of this Article VII shall be construed in accordance with all applicable Code sections. As a nonqualified deferred compensation plan described in Code Section 457(b), distributions under this Plan are generally includible in income for income tax purposes at the time such distributions are received.

(b) <u>Direct Rollover Option</u>

(1) In General

A Participant or Beneficiary who is otherwise entitled to a distribution under this Section 7.02 as a Distributee may elect, at the time and in the manner determined by the Plan Administrator, to have part or all of the distribution paid in a Direct Rollover to an Eligible Retirement Plan specified by the Participant or Beneficiary in accordance with Code Section 457(e)(16), provided the amount to be rolled over is an Eligible Rollover Distribution that meets one of the following minimum requirements:

- (A) <u>Total Account</u>. If the amount of the Direct Rollover is the total amount of the interest of the Participant or Beneficiary in the Plan, then such amount must be at least \$200 (or such other minimum amount required under the Code or Treasury Regulations).
- (B) <u>All of Distribution</u>. If the amount of the Direct Rollover is the full amount of the distribution, then it must be reasonable to expect that the total amount of all distributions to the Participant or Beneficiary for the calendar year will be at least \$200 (or such other minimum amount required under the Code or Treasury Regulations).

(C) <u>Part of Distribution</u>. If the amount of the Direct Rollover is part, but not all of the distribution, then such amount is at least \$500 (or such other minimum amount required under the Code or Treasury Regulations).

(2) <u>Definitions</u>

For purposes of this Section 7.02 and Section 7.04(d), the following definitions shall apply:

- (A) "Eligible Rollover Distribution" shall mean any distribution of all or any portion of the balance to the credit of the Distributee, except that an Eligible Rollover Distribution shall not include: (1) any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the Distributee or the joint lives (or joint life expectancies) of the Distributee and the Distributee's beneficiary, or for a specified period of ten years or more; (2) any distribution to the extent such distribution is required under Code Section 401(a)(9); (3) the portion of any distribution that is not includible in gross income (determined without regard to the exclusion for net unrealized appreciation with respect to employer securities); and (4) the portion of any distribution that is a hardship distribution.
- (B) "Eligible Retirement Plan" shall mean any of the following accounts or plans to the extent it accepts the Distributee's Eligible Rollover Distribution, except as otherwise provided in the next paragraph: (1) an individual retirement account described in Code Section 408(a); (2) an individual retirement annuity described in Code Section 408(b); (3) an annuity plan described in Code Section 403(a); (4) an annuity contract described in Code Section 403(b); (5) an eligible plan under Code Section 457(b) which is maintained by a state, political subdivision of a state, or an agency or instrumentality of a state or political subdivision of a state, and which agrees to separately account for amounts transferred into such plan from this Plan; (6) a qualified retirement plan described in Code Section 401(a); and (7) a Roth IRA described in Code Section 408A. The preceding definition of Eligible Retirement Plan shall also apply in the case of an Eligible Rollover Distribution to the Participant's Spouse, surviving Spouse, or former Spouse who is the Alternate Payee under a Qualified Domestic Relations Order.

However, for the portion, if any, of the Eligible Rollover Distribution that consists of after-tax employee contributions which are not includible in gross income (other than designated Roth contributions under Code Section 402A), "Eligible Retirement Plan" shall mean any of the following accounts or plans to the extent it accepts the Distributee's Eligible Rollover Distribution and agrees to separately account for

amounts so transferred, including separately accounting for the portion of such distribution which is includible in gross income and the portion of such distribution which is not includible: (1) an individual retirement account described in Code Section 408(a); (2) an individual retirement annuity described in Code Section 408(b); (3) an annuity plan described in Code Section 403(a); (4) a qualified retirement plan described in Code Section 401(a); and (5) for distributions after December 31, 2006, an annuity plan described in Code Section 403(b). Moreover, for any portion of the Eligible Rollover Distribution that is attributable to payments from a Roth Contribution Account, an Eligible Retirement Plan with respect to such portion shall mean only: a Roth IRA described in Code Section 408A; a designated Roth account described in Code Section 402A that is under a qualified Code Section 401(k) retirement plan; and a designated Roth account described in Code Section 402A that is under a Code Section 403(b) annuity plan.

- (C) "Distributee" shall include an Employee or former Employee. In addition, the Employee's Spouse or former Spouse who is the alternate payee under a Qualified Domestic Relations Order shall be treated as a Distributee with respect to the interest of such Spouse or former Spouse. Moreover, effective January 1, 2010, a non-Spouse Beneficiary who is a designated beneficiary under Code Section 401(a)(9)(E) and Treasury Regulation Section 1.401(a)(9)-4, may also be a Distributee for purposes of electing a Direct Rollover of an Eligible Rollover Distribution under Section 7.02 and Section 7.04(d), but only if the Eligible Retirement Plan accepting the Direct Rollover is an individual retirement account or an individual retirement annuity.
- (D) "<u>Direct Rollover</u>" shall mean a payment by the Plan to the Eligible Retirement Plan specified by the Distributee.
- (c) [Reserved]
- (d) [Reserved]
- (e) [Reserved]
- (f) Special Rule for Roth Contribution Accounts

Generally, in accordance with Code Section 402A, a distribution from a Participant's Roth Contribution Account (comprising Roth Contributions plus earnings) would constitute a "qualified Roth distribution" ("Qualified Roth Distribution"), and would not be includible in income of the recipient, only if: (1) the Roth Contribution Account has been maintained for five taxable years of the Participant beginning with the first day of the taxable year of the Participant in which the Participant makes Roth Contributions to such Account (or, in the case of a rollover contribution to the Roth Contribution Account from a previously established Roth contribution account under

another qualified Code Section 401(k) retirement plan, for five taxable years of the Participant beginning with the first day of the taxable year of the Participant in which the Participant makes Roth contributions to such previously established Roth contribution account); and (2) the distribution is made after the Participant attains age 59½, the Participant becomes disabled, or the Participant dies. A distribution of Excess Elective Employer Contributions or Excess Deferrals would not constitute a Qualified Roth Distribution.

- (1) Order of Priority. Unless otherwise provided under the Plan, if a Participant's Plan Account includes a Roth Contribution Account, then any distribution of the Participant's Plan Account shall be made in the following order of priority: (A) first, as a distribution of the Account that is not comprised of the Roth Contribution Account; and (B) second, as a distribution of the Account that is comprised of the Roth Contribution Account. Therefore, the Roth Contribution Account shall be distributed only after the distribution of the non-Roth Contribution Account.
- (2) <u>Exception for Direct Rollover</u>. To the extent that a Roth Contribution Account becomes distributable as an Eligible Rollover Distribution, the Participant shall be allowed to elect a Direct Rollover of his Roth Contribution Account, notwithstanding that such Direct Rollover may precede the distribution of the non-Roth Contribution Account.
- (3) <u>In-Plan Roth Conversion Amounts</u>. In addition to those conditions described above in this Section 7.02(f), a Participant may withdraw In-Plan Roth Conversion amounts, as described in Section 4.01(c)(1), from the Participant's Roth Contribution Account only when the Participant is eligible for a distribution from the Plan account that is the source of the In-Plan Roth Conversion amount. An In-Plan Roth Conversion shall not expand or eliminate any distribution rights, including any optional forms of benefit described in Code Section 411(d)(6), or any restrictions on amounts that a Participant elects to treat as an In-Plan Roth Conversion.

7.03 <u>Time Limits for Commencing Distributions</u>

(a) General Distribution Restrictions

Distribution of a Participant's Account shall be made no earlier than (1) severance from Service in accordance with Section 7.01; (2) the calendar year in which the Participant attains age 70½; (3) in the event of an approved financial hardship due to an "Unforeseeable Emergency" in accordance with Section 7.05; or (4) a Participant's Account meets the requirements for an in-service distribution in accordance with Section 7.06.

(b) Required Beginning Date under Code Section 401(a)(9)

Notwithstanding any provision of the Plan to the contrary, a Participant's vested interest in his Account shall be distributed or begun to be distributed no later than

the Participant's "Required Beginning Date" and shall be made in accordance with the requirements described in this Section 7.03(b) and shall otherwise comply with requirements of Code Section 401(a)(9) and underlying Treasury Regulations. Except as provided in the next paragraph, a Participant's "Required Beginning Date" shall be April 1 of the calendar year following the later of (1) the calendar year in which the Participant attains age 70½ or (2) the calendar year in which the Participant retires.

However, in the case of a Participant who is a 5% Owner at any time during the five Plan Year period ending in the calendar year in which the Participant attains age 70½, the Required Beginning Date shall be April 1, of the calendar year following the calendar year in which he attains age 70½. Further, in the case of a Participant who becomes a 5% Owner during any subsequent Plan Year, the Required Beginning Date shall be the April 1 of the calendar year following the calendar year in which such subsequent Plan Year ends. Once distributions have commenced to a Participant who is a 5% Owner, the distributions shall continue even if the Participant ceases to be a 5% Owner in a subsequent year.

(c) <u>Minimum Distribution Requirements</u>

(1) General Rules

- (A) <u>Effective Date</u>. The provisions of this Section 7.03(c) shall apply for purposes of determining required minimum distributions for calendar years beginning with the 2003 calendar year.
- (B) <u>Precedence</u>. The requirements of this Section 7.03(c) shall take precedence over any inconsistent provisions of the Plan.
- (C) Requirements of Treasury Regulations. All distributions required under this Section 7.03(c) shall be determined and made in accordance with the Treasury Regulations under Section 401(a)(9) of the Code and the minimum distribution incidental benefit requirement of Code Section 401(a)(9)(G), which are incorporated herein by this reference.

(2) Time and Manner of Distribution

- (A) <u>To Participant</u>. The Participant's entire interest shall be distributed, or begin to be distributed, to the Participant no later than the Participant's Required Beginning Date.
- (B) <u>To Beneficiary</u>. If the Participant dies before distributions begin, the Participant's entire interest shall be distributed, or begin to be distributed, no later than as follows:
 - (i) <u>Surviving Spouse</u>. If the Participant's surviving Spouse is the Participant's sole designated Beneficiary, distributions to the surviving Spouse shall begin by December 31

of the calendar year immediately following the calendar year in which the Participant died, or by December 31 of the calendar year in which the Participant would have attained age 70 ½, if later. The beginning date shall not be required if the surviving Spouse elects instead to complete distributions by December 31 of the calendar year containing the fifth anniversary of the Participant's death. The election must be made no later than September 30 of the earlier of the calendar year in which distributions would otherwise be required to begin if the election were not made, or the calendar year which contains the fifth anniversary of the Participant's (or, if applicable, the surviving Spouse's) death.

- (ii) Other Designated Beneficiary. If the Participant's surviving Spouse is not the Participant's sole designated Beneficiary, distributions to the designated Beneficiary shall begin by December 31 of the calendar year immediately following the calendar year in which the Participant died. The beginning date shall not be required if the designated Beneficiary elects instead to complete distributions by December 31 of the calendar year containing the fifth anniversary of the Participant's death. The election must be made no later than September 30 of the earlier of the calendar year in which distributions would otherwise be required to begin if the election were not made, or the calendar year which contains the fifth anniversary of the Participants (or, if applicable, the surviving Spouse's) death.
- (iii) <u>No Designated Beneficiary</u>. If there is no designated Beneficiary as of September 30 of the year following the year of the Participant's death, the Participant's entire interest shall be distributed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
- (iv) <u>Death of Surviving Spouse</u>. If the Participant's surviving Spouse is the Participant's sole designated Beneficiary and the surviving Spouse dies after the Participant but before distributions to the surviving Spouse begin, this Section 7.03(e)(2)(B), other than Section 7.03(c)(2)(B)(i), shall apply as if the surviving Spouse were the Participant.

For purposes of this Section 7.03(c)(2)(B) and Section 7.03(c)(4), unless Section 7.03(c)(2)(B)(iv) applies, distributions are considered to begin on the Participant's Required Beginning Date. If Section 7.03(c)(2)(B)(iv) applies, distributions are considered to begin on the date distributions are required to begin to the surviving Spouse under Section 7.03(c)(2)(B)(i). If distributions under an annuity purchased from an insurance company irrevocably commence to the Participant before the Participant's Required Beginning Date (or to the

Participant's surviving Spouse before the date distributions are required to begin to the surviving Spouse under Section 7.03(c)(2)(B)(i)), the date distributions are considered to begin is the date distributions actually commence.

(C) Forms of Distribution. Unless the Participant's interest is distributed in the form of an annuity purchased from an insurance company or in a single sum on or before the Required Beginning Date, as of the first distribution calendar year, distributions shall be made in accordance with Section 7.03(c)(3) and Section 7.03(c)(4) of this Article. If the Participant's interest is distributed in the form of an annuity purchased from an insurance company, distributions thereunder shall be made in accordance with the requirements of Section 401(a)(9) of the Code and Treasury Regulations.

(3) Required Minimum Distributions during Participant's Lifetime

During the Participant's lifetime, the minimum amount distributed for each Distribution Calendar Year (as defined in Section 7.03(c)(5)(B) shall be the lesser of:

- (A) the quotient obtained by dividing the Participant's Account balance by the distribution period in the Uniform Lifetime Table set forth in Section 1.401(a)(9)-9 of the Treasury Regulations, using the Participant's age as of the Participant's birthday in the Distribution Calendar Year (as defined in Section 7.03(c)(5)(B)); or
- (B) if the Participant's sole designated Beneficiary for the Distribution Calendar Year is the Participant's Spouse, the quotient obtained by dividing the Participant's Account balance by the number in the Joint and Last Survivor Table set forth in Section 1.401(a)(9)-9 of the Treasury Regulations, using the Participant's and Spouse's attained ages as of the Participant's and Spouse's birthdays in the Distribution Calendar Year.

Required minimum distributions shall be determined under this Section 7.03(c)(3) beginning with the first Distribution Calendar Year and up to and including the Distribution Calendar Year that includes the Participant's date of death.

(4) Required Minimum Distributions After Participant's Death

(A) <u>Death on or after Date Distributions Begin</u>

(i) <u>Participant Survived by Designated Beneficiary</u>. If the Participant dies on or after the date distributions begin and there is a designated Beneficiary, the minimum amount that shall be distributed for each Distribution Calendar Year after the year of

the Participant's death shall be the quotient obtained by dividing the Participant's Account balance by the longer of the remaining life expectancy of the Participant or the remaining life expectancy of the Participant's designated Beneficiary, determined as follows:

- (I) The Participant's remaining life expectancy shall be calculated using the age of the Participant in the year of death, reduced by one for each subsequent year.
- (II) If the Participant's surviving Spouse is the Participant's sole designated Beneficiary, the surviving Spouse's remaining life expectancy shall be calculated for each Distribution Calendar Year after the year of the Participant's death using the surviving Spouse's age as of the Spouse's birthday in that year. For Distribution Calendar Years after the year of the surviving Spouse's death, the remaining life expectancy of the surviving Spouse is calculated using the age of the surviving Spouse as of the Spouse's birthday in the calendar year of the Spouse's death, reduced by one for each subsequent calendar year.
- (III) If the Participant's surviving Spouse is not the Participant's sole designated Beneficiary, the designated Beneficiary's remaining life expectancy shall be calculated using the age of Beneficiary in the year following the year of the Participant's death, reduced by one for each subsequent year.
- (ii) No Designated Beneficiary. If the Participant dies on or after the date distributions begin and there is no designated Beneficiary as of September 30 of the year after the year of the Participant's death, the minimum amount that shall be distributed for each distribution calendar year after the year of the Participant's death shall be the quotient obtained by dividing the Participant's Account balance by the Participant's remaining life expectancy calculated using the age of the Participant in the year of death, reduced by one for each subsequent year.

(B) <u>Death before Date Distributions Begin</u>

(i) <u>Designated Beneficiary</u>. If the Participant dies before the date distributions begin and there is a designated Beneficiary, the minimum amount that shall be distributed for each Distribution Calendar Year after the year of the Participant's death shall be the quotient obtained by dividing the Participant's Account balance by the remaining life expectancy of the Participant's

designated Beneficiary, determined as provided in Section 7.03(c)(4)(A).

- (ii) <u>No Designated Beneficiary</u>. If the Participant dies before the date distributions begin and there is no designated Beneficiary as of September 30 of the year following the year of the Participant's death, distribution of the Participant's entire interest shall be completed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
- (iii) Death of Surviving Spouse Before Distributions to Surviving Spouse Are Required to Begin. If the Participant dies before the date distributions begin, and the Participant's surviving Spouse is the Participant's sole designated Beneficiary, and if the surviving Spouse dies before distributions are required to begin to the surviving Spouse under Section 7.03(c)(2)(B)(i), then this Section 7.03(c)(4)(B) shall apply as if the surviving Spouse were the Participant.

(5) Definitions

- (A) "Designated Beneficiary" shall mean the individual who is designated as the Beneficiary in accordance with Section 1.04 of the Plan and is the designated Beneficiary under Section 401(a)(9) of the Internal Revenue Code and Section 1.401(a)(9)-4, of the Treasury Regulations. Section 1.401(a)(9)-4, Q&A-5, of the Treasury Regulations permits the designation of a trust as Beneficiary provided the requirements of the cited Section are met.
- (B) "Distribution Calendar Year" shall mean a calendar year for which a minimum distribution is required. For distributions beginning before the Participant's death, the first Distribution Calendar Year shall be the calendar year immediately preceding the calendar year which contains the Participant's Required Beginning Date. For distributions beginning after the Participant's death, the first Distribution Calendar Year shall be the calendar year in which distributions are required to begin under Section 7.03(c)(2)(B). The required minimum distribution for the Participant's first Distribution Calendar Year shall be made on or before the Participant's Required Beginning Date. The required minimum distribution for other Distribution Calendar Years, including the required minimum distribution for the distribution calendar year in which the Participant's Required Beginning Date occurs, shall be made on or before December 31 of that Distribution Calendar Year.
- (C) "<u>Life Expectancy</u>" shall mean life expectancy as computed by use of the Single Life Table in Section 1.401(a)(9)-9 of the Treasury Regulations.

- (D) "Participant's Account Balance" shall mean the Account balance as of the last Valuation Date in the calendar year immediately preceding the Distribution Calendar Year (valuation calendar year) increased by the amount of any contributions made and allocated or forfeitures allocated to the Account balance as of dates in the valuation calendar year after the Valuation Date and decreased by distributions made in the valuation calendar year after the Valuation Date. The Account balance for the valuation calendar year includes any amounts rolled over or transferred to the Plan either in the valuation calendar year or in the distribution calendar year if distributed or transferred in the valuation calendar year.
- (E) "Required Beginning Date" shall mean the date specified in Section 7.03(b) of the Plan.

7.04 Administration of Distributions

(a) In General

Benefits under the Plan shall be distributed solely from the Trust. No distributions shall be made or commence prior to the Participant's termination from Service, except as required under Article VII and Section 11.02, and in the case of termination of the Plan.

(b) Authority

Distributions shall be made only in accordance with the directions of the Plan Administrator. The Plan Administrator shall have the authority to direct the distributions in accordance with the terms and conditions of the Plan, but the Plan Administrator shall have no power of discretionary consent with regard to a Participant's or a Beneficiary's choice of the form or timing of a distribution, except as specifically stated herein, to the extent that the Plan Administrator is constrained by the options available under the Plan or by the requirements of law or regulation.

The Plan Administrator may determine that a Participant's election as to the timing and form of his distribution shall not be treated as valid unless the election is completed on a form prescribed or approved by the Plan Administrator, and unless such completed form is received and acknowledged by the Plan representatives.

(c) Minors or Legally Incompetents

In the case of any distribution to a minor or to a legally incompetent person, the Plan Administrator may: (1) direct the distribution to his legal representative, to a designated relative, or directly to such person for his benefit; or (2) instruct the distribution directly for his support, maintenance, or education.

(d) Written Explanations Regarding Eligible Rollover Distributions

(1) Required Tax Information

The Plan Administrator shall provide a Distributee (as described in Section 7.02(b)(2)(C) who is to receive an Eligible Rollover Distribution, with a "Section 402(f) Notice" not less than 30 days nor more than 180 days before making the distribution in accordance with Code Section 457(b)(16)(B). A "Section 402(f) Notice" shall be a written explanation of the various rules allowing the Distributee to elect a direct rollover to an Eligible Retirement Plan, requiring the withholding of tax on the distribution if it is not paid out in a direct rollover, allowing deferral of tax if a rollover is made within 60 days of a distribution, and other rules regarding the taxation of an Eligible Rollover Distribution, as described in Code Section 402(f).

(e) Explanation of a Right to Defer

Distribution forms issued in Plan Years beginning after December 31, 2006, shall include a description of a Participant's right, if any, to defer receipt of a distribution and shall describe the consequences of failing to defer receipt of the distribution. Subject to Internal Revenue Service guidance, the distribution forms shall include: (1) a description of the federal tax implications of failing to defer or to rollover a distribution; (2) a description indicating the investment options available under the Plan (including fees) that are available if the Participant defers distribution; and (3) a reference to the portion of the summary plan description that contains any special rules that might materially affect a Participant's decision to defer.

7.05 <u>In-Service Withdrawals for Hardship</u>

At the request of a Participant, the Plan Administrator may authorize a withdrawal at any time from the Participant's Account, provided that the authorization for such withdrawal and the amount thereof shall be given only on account of hardship incurred by the Participant due to an "Unforeseeable Emergency".

For this purpose, an "Unforeseeable Emergency" shall mean a severe financial hardship to the Participant resulting from: (a) a sudden and unexpected illness or accident of the Participant, the Participant's Beneficiary, or the Participant's or Beneficiary's Spouse or dependent (within the meaning of Code Section 152); (b) loss of the Participant's or Beneficiary's property or due to casualty; (c) the need to pay for the funeral expenses of the Participant's or Beneficiary's Spouse or dependent (within the meaning of Code Section 152); or (d) other similar extraordinary and unforeseeable circumstances arising as a result of events beyond the Participant's or Beneficiary's control. The circumstances that will constitute an Unforeseeable Emergency shall depend upon the facts and circumstances of each case, but in any case, payment may not be made to the extent that such hardship is or may be relieved: (a) through reimbursement or compensation by insurance or otherwise; (b) by distribution of the Participant's assets, to the extent that distribution of such assets would not itself cause severe financial hardship; or (c) by cessation of deferrals under the Plan. The need to send a Participant's child to

college or the desire to purchase a home shall not be considered to be an Unforeseeable Emergency.

7.06 <u>In-Service Distributions</u>

- (a) <u>Voluntary In-Service Distribution of De Minimis Accounts</u>. The Plan Administrator may permit a Participant, while in employment with the Employer, to elect an in-service distribution of the total amount payable to the Participant under the Plan if the following requirements are met:
 - (1) The total amount payable to the Participant under the Plan excluding Rollover Contributions does not exceed an amount specified from time to time by the Plan Administrator (and not in excess of \$5,000 or other applicable limit under the Code Section 411(a)(11)(A));
 - (2) The Participant has not previously received an in-service distribution of the total amount payable to the Participant under the Plan; and
 - (3) No amount has been deferred under the Plan with respect to the Participant during the two-year period ending on the date of the in-service distribution.

7.07 Plan to Plan Transfers

(a) Post-Severance Transfer

At the sole discretion of the Plan Administrator, and subject to the requirements of this Section 7.07(a), the Plan may upon request transfer the Account balance of a Participant or Beneficiary to another eligible governmental plan within the meaning of Code Section 457(b) and Treasury Regulation Section 1.457-2(f) that is maintained by another employer, or accept a transfer of an account from such a plan.

Such transfer may only be allowed if the transferor plan provides for transfers, the receiving plan provides for the receipt of transfers, the participant or beneficiary whose amounts deferred are being transferred will have an amount deferred immediately after the transfer at least equal to the amount deferred with respect to that participant or beneficiary immediately before the transfer, and in the case of a transfer for a participant, the participant has had a severance from employment with the transferring employer and is performing services for the entity maintaining the receiving plan.

The transferred amounts shall be subject to distribution restrictions of Treasury Regulation Section 1.457-6 in the receiving plan in the same manner as if the transferred amount had been originally been deferred under the receiving plan if the participant is performing services for the entity maintaining the receiving plan. Such transfer shall be carried out in accordance with the requirements of Treasury Regulation Section 1.457-10(b).

(b) <u>Transfer Among Plans of the Employer</u>

At the sole discretion of the Plan Administrator, and subject to the requirements of this Section 7.07(b), the Plan may upon request transfer the Account balance of a Participant or Beneficiary to another eligible governmental plan within the meaning of Code Section 457(b) and Treasury Regulation Section 1.457-2(f) that is maintained by the Employer, or accept a transfer of an account from such a plan.

Such transfer may only be allowed if both the transferor and receiving plans are sponsored by the Employer (and, for this purpose, the Employer is not treated as the Employer if the participant's compensation is paid by a different entity), the transferor plan provides for transfers, the receiving plan provides for the receipt of transfers, the participant or beneficiary whose amounts deferred are being transferred will have an amount deferred immediately after the transfer at least equal to the amount deferred with respect to that participant or beneficiary immediately before the transfer, and the participant or beneficiary whose deferred amounts are being transferred is not eligible for additional annual deferrals in the receiving plan unless the participant or beneficiary is performing services for the entity maintaining the receiving plan.

The transferred amounts shall be subject to distribution restrictions of Treasury Regulation Section 1.457-6 in the receiving plan in the same manner as if the transferred amount had been originally been deferred under the receiving plan if the participant is performing services for the entity maintaining the receiving plan. Such transfer shall be carried out in accordance with the requirements of Treasury Regulation Section 1.457-10(b).

(c) <u>Permissive Service Credit Transfers</u>

If a Participant is also a participant in a tax-qualified defined benefit governmental plan (as defined in Code Section 414(d)) that provides for the acceptance of transfers with respect to the Participant, the Plan Administrator may in its sole discretion, permit a Participant to elect to have any portion of the Participant's Account transferred to such defined benefit governmental plan. A transfer under this Section 7.07(c) shall not be treated as a distribution for purposes of Treasury Regulation Section 1.457-6 and may be made before the Participant has had a severance from Service.

A transfer may be made under this Section 7.07(c) if the transfer is either for the purchase of permissive service credit (as defined in Code Section 415(n)(3)(A)) under the receiving defined benefit governmental plan or a repayment to which Code Section 415 does not apply by reason of Code Section 415(k)(3) or as otherwise allowed by the Internal Revenue Service.

ARTICLE VIII

DESIGNATION OF BENEFICIARY

8.01 Designation Procedure

Each Employee, upon becoming a Participant in the Plan, may designate a Beneficiary or Beneficiaries to receive the benefits provided under the Plan in the event of the Participant's death. A Participant shall be allowed to revoke any prior designation of Beneficiary and to change his Beneficiary designation at any time. Each Beneficiary designation shall be in a form prescribed by the Plan Administrator and shall be effective only when filed with, and acknowledged by, the Plan Administrator during the Participant's lifetime. Each Beneficiary designation filed with the Plan Administrator shall cancel any previously filed Beneficiary designations.

8.02 Lack of Designation

If it is determined by the Plan Administrator that all the individuals designated by the Participant as Beneficiaries to receive his interest in the Plan by reason of his death have predeceased the Participant or are otherwise incapable of receiving the interest, the Participant's interest shall be paid to the Participant's estate.

ARTICLE IX

ADMINISTRATION

9.01 Plan Governance

This Article describes the allocation of powers and duties for the operation of the Plan, Plan administration, and management of Plan assets among the Board of Trustees, the Plan Administrator, and the other parties, if any, involved in Plan Administration.

(a) Board of Trustees

In accordance with GCA Sections 8301(b) and 8304, the Board of Trustees shall be responsible for the direction and operation of the affairs and business of the Plan, the administration of the Plan, and may conduct all the business of the Plan.

As described in GCA Section 8305, the Board of Trustees shall further have the following discretionary authority in the administration of the Plan:

- (1) To adopt the Plan and any amendments which the Board of Trustees shall determine to be required in order that the Plan shall comply with the requirements of Code Section 457;
- (2) To exercise original jurisdiction in all matters relating to or affecting the Plan, including, but not limited to, claims for benefits and refunds;
- (3) To approve such rules, regulations, administrative procedures and prescribed forms as established by the Director, as the Board of Trustees may deem necessary to effectuate the purposes of the Plan;
- (4) To establish Accounts for each Participant, to which his respective contributions, together with all earnings on investments shall be deposited, and from which his respective benefits under the Plan and his respective pro-rata share of reasonable administrative fees and expenses, as authorized by the Board of Trustees, shall be paid;
- (5) To engage actuarial, legal, clerical and all other advisors, including, but not limited to, a Qualified Custodian, as may be necessary to effectuate the purposes of the Plan;
- (6) To cause to be maintained adequate accounting records that reflect the financial condition of the Plan, and such additional data as is necessary or required for its operation; and
- (7) To establish an annual budget for costs in the soliciting, evaluating and periodic review process of companies offering Investment Options, and to charge a proportional share of all costs related to the periodic review to each such company currently under contract and charge a proportional share of all costs related to soliciting and evaluating bids to each such company offering investment options which is selected by the Board of Trustees.

As described in GCA Section 8305, the Board of Trustees shall further have the following discretionary authority with respect to investments:

- (1) To enter into annuity contracts, trust agreements and custodial arrangements as required in order that the Plan shall comply with the requirements of Code Section 457(g);
- (2) To contract with an insurance, annuity, mutual fund, trust company or other qualified company to administer the operations of the Plan;
- (3) To take any and all additional actions, including, but not limited to, the execution of custodial, record keeping, and investment advisory agreements in order to implement the Plan;
- (4) To solicit bids for investment options and retain consulting services to assist it in soliciting and evaluating bids and in the periodic review of companies offering investment options;
- (5) To invest and reinvest the monies of the Plan and manage the assets and income of the Plan; and
- (6) To delegate its authority to invest or manage the assets and income of the Plan, provided that the Board of Trustees shall exercise reasonable care, skill and caution in selecting an agent, establishing the scope and terms of the delegation, consistent with the purposes and terms of the Plan, and periodically reviewing the agent's performance and compliance with the terms of the delegation.

(b) <u>Plan Administrator</u>

The Director shall be the Plan Administrator. As described in GCA Section 8305, the Director shall establish rules, regulations, administrative procedures and prescribed forms for the administration of the Plan. As Plan Administrator, the Director shall further have responsibility and discretionary authority for making decisions regarding Plan administration and Plan investments, other than any decisions falling within the scope of the Board of Trustees' sole responsibilities, to the extent such responsibilities and discretionary authority have been delegated to the Plan Administrator by the Board of Trustees.

9.02 Plan Representatives

The Board of Trustees or the Plan Administrator may appoint Plan Representatives to represent the Plan, the Board of Trustees, or the Plan Administrator in the day-to-day administration of the Plan.

9.03 Reliance

To the extent permitted by law, the Board of Trustees, the Plan Administrator and any person to whom either may delegate any duty or power in connection with administering the Plan, the Employer, and the officers and directors thereof, shall be entitled to rely conclusively upon, and shall be fully protected in any action taken or suffered by them in good faith in the

reliance upon, an enrolled actuary, legal counsel, accountant, other specialist, or other person selected by the Board of Trustees or the Plan Administrator, or in reliance upon any tables, valuations, certificates, opinions or reports which shall be furnished by any of them. Further, to the extent permitted by law, no member of a committee, nor the Employer, nor the officers nor directors thereof, shall be liable for any neglect, omission or wrongdoing of any other member of the Board of Trustees or the Plan Administrator.

9.04 Indemnification

The Employer may, in its discretion, purchase insurance to protect the Board of Trustees, the Plan Administrator, or Plan Representatives against any potential liability under the terms of this Plan. The Employer shall indemnify the members of the Board of Trustees, the Plan Administrator, and Plan Representatives against all claims and demands for loss or damage and from and against all losses and expenses, including reasonable attorney's fees, arising out of or in connection with the performance of their obligations under this Plan, provided that any acts or omissions which are the basis for such claims, demands, losses and expenses are not the result of willful misconduct of the person being indemnified.

9.05 Expenses

All usual and reasonable expenses that shall arise in connection with the administration of the Plan, including but not limited to, administrative expenses and proper charges and disbursements under the Plan, and compensation and other expenses and charges of any legal counsel, enrolled actuary, accountant, specialist, or other person who shall be employed by the Board of Trustees or the Plan Administrator in connection with the administration thereof, shall be paid from the assets credited under the Plan to the extent not otherwise paid by the Employer, in accordance with GCA Section 8302(a). Any payment of such expenses shall be in accordance with GCA Section 8307.

9.06 Claims for Benefits

Any Participant, surviving Spouse, or other Beneficiary making a claim for benefits under the Plan shall file a written application with the Plan Administrator. In the event of a mistake or misstatement as to the eligibility, participation, or service of any Participant or the amount of payments made or to be made to a Participant, or Beneficiary, the Plan Administrator shall, if possible, cause to be withheld or accelerated or otherwise make adjustment of the amounts of payments as will in its sole judgment result in the Participant or Beneficiary receiving the proper amount of payments under the Plan.

ARTICLE X

INVESTMENTS

10.01 Funding

The Plan shall be funded on a current basis from Employer Contributions in accordance with GCA Section 8303. All amounts of Compensation deferred pursuant to the Plan, all property and rights purchased with such amounts and all income attributable to such amounts, property or rights shall be held in Trust by the Trustee for the exclusive benefit of Participants and their Beneficiaries, in accordance with Code Section 457(g), Treasury Regulation Section 1.457-8(a), and GCA Sections 8301(b), 8302(a), and 8306.

The Trust shall be established pursuant to a written agreement that constitutes a valid trust under applicable state law. It shall be impossible, prior to the satisfaction of all liabilities with respect to Participants and their Beneficiaries, for any part of the assets and income of the Trust to be used for, or diverted to, purposes other than for the exclusive benefit of Participants and their Beneficiaries. All amounts of Compensation deferred pursuant to the Plan shall be transferred to the Trust within a period that is not longer than is reasonable for the proper administration of the accounts of Participants.

For purposes of this Section 10.01, a custodial account shall be treated as the Trust or part thereof if the custodian of such custodial account is a Qualified Custodian. Compensation deferred under the Plan which is to be held in such a custodial account shall be transferred to the custodial account within a period that is not longer than is reasonable for the proper administration of the Accounts of Participants, and such custodial account shall be a custodial account as described in Code Section 401(f).

For purposes of this Section 10.01, an annuity contract shall be treated as the Trust or part thereof. Such annuity contract shall be an annuity as defined in Code Section 401(g), be issued by an insurance company qualified to do business in the state where the contract was issued, and be for the exclusive benefit of Participants and Beneficiaries under the Plan. Such annuity contract shall not be a life, health or accident, property, casualty, or liability insurance contract. Compensation deferred under the Plan which is to be held in such an annuity contract shall be invested within a period that is not longer than is reasonable for the proper administration of the Accounts of Participants. Such annuity contract shall be an annuity contract as described in Code Section 401(f).

The trust requirement of this Section 10.01 may be satisfied through use of a trust agreement, one or more custodial accounts, or one or more annuity contracts, as described herein, or a combination thereof, in accordance with Treasury Regulation Section 1.457-8(a).

10.02 Separate Investment Funds

(a) Plan Administrator May Establish Investment Options

The Plan Administrator may, in its sole discretion, create one or more separate Investment Options having such different specific investment objective as the

Plan Administrator shall from time to time determine, in accordance with GCA Sections 8305(e) and 8306. The Plan Administrator shall determine and may from time to time redetermine the number of Investment Options and the specific objectives of the Investment Options. From time to time the Plan Administrator may add or delete Investment Options without amending the Plan. Participants shall be informed of the various Investment Options available.

(b) Participant Direction Permitted

The Plan Administrator may, in its sole discretion, allow each Participant the right to instruct the Trustee in writing to invest his Account in one or more separate Investment Options. In the event a Participant fails to direct the investment of his Account at such time and in such manner as the Plan Administrator may require, the Participant shall be presumed to have elected to direct the investment of his Account as well as any future contributions and earnings to be allocated to his Account in the default Investment Option which the Plan Administrator may designate from time to time. Finally, the Plan Administrator may, but is not required to, permit Beneficiaries of deceased Participants to direct the investment of an Account in which they have an interest.

(c) Plan Administrator to Establish Rules

The Plan Administrator may at any time make such rules as it determines necessary regarding the administration of a directed Investment Option described in this Section 10.02. The Plan Administrator shall develop and maintain rules governing the rights of Participants to change their investment directions and the frequency with which such changes can be made.

10.03 Loans To Participants

A Participant who is an Employee may apply for and receive a loan from his or her Account balance as provided in this Section 10.03 or pursuant to a loan policy executed by the Plan Administrator. Any such loan may not be for an amount less than the minimum amount specified by the Plan Administrator. If not otherwise specified by the Administrator, the minimum loan amount shall be \$1,000.

Except as modified by the loan policy adopted by the Plan Administrator from time to time, the following rules shall apply to loans under the Plan. Any loans that are issued under the Plan shall be administered in a manner consistent with the requirements contained in Code Section 72(p), Treasury Regulation Section 1.72(p)-1, and any applicable guidance issued thereunder.

(a) Maximum Loan Amount

No loan to a Participant hereunder may exceed the lesser of (1) \$50,000, reduced by the greater of (A) the outstanding balance on any loan from the Plan to the Participant on the date the loan is made or (B) the highest outstanding balance on loans from the Plan to the Participant during the one-year period ending on the day before the

date the loan is approved by the Plan Administrator (not taking into account any payments made during such one-year period), or (2) one-half of the value of the Participant's vested Account balance.

For purposes of this section 10.03, any loan from any other plan maintained by the Employer shall be treated as if it were a loan made from this Plan, and the Participant's vested interest under any such other plan shall be considered a vested interest under this Plan; provided, however, that the provisions of this paragraph shall not be applied so as to allow the amount of a loan under this section 10.03 to exceed the amount that would otherwise be permitted in the absence of this paragraph.

(b) <u>Loan Provisions</u>

The terms of any loan shall (1) require level amortization with payments not less frequently than quarterly throughout the repayment period, except that alternative arrangements for repayment may apply in the event that the borrower is on a bona fide unpaid leave of absence for a period not to exceed one year for leaves other than a qualified military leave within the meaning of Code Section 414(u) or for the duration of an interruption of employment which is due to qualified military service, (2) require that the loan be repaid within five years unless the Participant certifies in writing to the Plan Administrator that the loan is to be used to acquire any dwelling unit which within a reasonable time is to be used (determined at the time the loan is made) as a principal residence of the Participant, in which case the term of such loan shall be determined at the Plan Administrator's discretion, and (3) provide for a reasonable rate of interest to be fixed by the Plan Administrator from time to time. The Plan Administrator shall not discriminate among Participants in the matter of interest rates, but loans granted at different times may bear different interest rates based upon prevailing rates at the time.

A loan to a Participant shall be considered a directed investment option for such Participant's account balance.

(c) <u>Security for Loan; Default</u>

- (1) <u>Security</u>. Any loan to a Participant under the Plan shall be secured by the pledge of the portion of the Participant's interest in the Plan invested in such loan.
- (2) <u>Default</u>. In the event that a Participant fails to make a loan payment under this Section 10.03 by the end of the calendar quarter following the calendar quarter in which such payment was due, a default on the loan shall occur. In the event of such default (A) all remaining payments on the loan shall be immediately due and payable, (B) interest will continue to accrue on the unpaid balance until the loan is repaid in full, and (C) the Participant shall be permanently ineligible for any future loans from the Plan unless, in the Plan Administrator's sole discretion, the Participant is deemed to be credit worthy and agrees to repay the loan through payroll deduction.

In the case of any default on a loan to a Participant, the Plan Administrator shall apply the portion of the Participant's interest in the Plan held as security for the loan in satisfaction of the loan on the date of severance from Service. In addition, the Plan Administrator shall take any legal action it shall consider necessary or appropriate to enforce collection of the unpaid loan, with the costs of any legal proceeding or collection to be charged to the Account balance of the Participant.

Notwithstanding any other provision of the Plan to the contrary, in the event a loan is outstanding hereunder on the date of a Participant's death, his or her estate shall be his or her Beneficiary as to the portion of his or her interest in the Plan invested in such loan (with the Beneficiary or Beneficiaries as to the remainder of his or her interest in the Plan to be determined in accordance with otherwise applicable provisions of the Plan).

(d) Repayment

The Participant shall be required, as a condition to receiving a loan, to enter into an irrevocable agreement authorizing the Employer to make payroll deductions from his or her Compensation so long as the Participant is an Employee and to transfer such payroll deduction amounts to the Trustee in payment of such loan plus interest. Repayments of a loan shall be made by payroll deduction of equal amounts (comprised of both principal and interest) from each paycheck, with the first such deduction to be made as soon as practicable after the loan funds are disbursed; provided, however, a Participant may prepay the entire outstanding balance of his loan at any time; and provided, further, that if any payroll deductions cannot be made in full because a Participant is on an unpaid leave of absence or is no longer employed by the Employer or the Participant's paycheck is insufficient for any other reason, the Participant shall pay directly to the Plan the full amount that would have been deducted from the Participant's paycheck, with such payment to be made by the last business day of the calendar month in which the amount would have been deducted. Loan repayments may, at the Plan Administrator's discretion, be suspended for Qualified Military Service as permitted by Code Section 414(u)(4), or an approved leave of absence as described by Treasury Regulation Section 1.72(p)-1.

ARTICLE XI

ASSIGNMENTS

11.01 No Assignment

Except as otherwise provided in Section 11.02 below, in accordance with GCA Section 8308(c), the interest herein of any Participant, former Participant, or Beneficiary, shall not be subject to alienation, assignment, pledging, encumbrance, attachment, garnishment, execution, sequestration, or other legal or equitable process, or transferability by operation of law in the event of bankruptcy, insolvency, or otherwise.

11.02 Qualified Domestic Relations Order

The provisions of this Article XI shall not prevent the creation, assignment, or recognition of any individual's right to a benefit payable with respect to a Participant pursuant to a Qualified Domestic Relations Order ("QDRO"). The Plan Administrator shall allow payments under a QDRO pursuant to the terms of the QDRO.

(a) QDRO Determination

The Plan Administrator shall establish reasonable, timely procedures to (1) determine whether a domestic relations order is a QDRO, (2) notify affected parties as specified in Code Section 414(p)(6), and (3) administer Plan distributions in accordance with a QDRO.

(b) Separate Accounting of Alternate Payee's Account

During any period in which the issue of whether a domestic relations order is a QDRO is being determined by the Plan Administrator, a court of law, or otherwise, the Plan Administrator may separately account for the amounts (with investment income and loss) which are involved.

(c) Consent Requirements

Except as otherwise provided in a QDRO, payments made to an Alternate Payee shall not be subject to (1) spousal consent, or (2) consent of the Alternate Payee.

(d) Payments May Occur Before Termination From Service

This Plan specifically permits distribution to an Alternate Payee under a QDRO at any time, including prior to the Participant's termination from Service, and irrespective of whether the Participant has attained his "earliest retirement age" (as defined in Code Section 414(p)) under the Plan. A distribution to an Alternate Payee prior to the Participant's attainment of earliest retirement age shall be allowed only if the order specifically provides for the distribution at such time.

ARTICLE XII

AMENDMENT AND TERMINATION

12.01 Amendment of Plan

The Board of Trustees reserves the right to amend the Plan to any extent and in any manner that it may deem advisable. Further, to the extent such authority has been delegated to the Plan Administrator by the Board of Trustees, the Plan Administrator may also amend the Plan. The Employer, all Participants, their Beneficiaries, and all other persons having any interest hereunder shall be bound by any such amendment, provided, however, that no amendment shall reduce the then accrued benefit of any Participant without the Participant's consent.

12.02 Termination of Plan

The Government of Guam has established the Plan with the bona fide intention and expectation that the Plan shall continue indefinitely, and that it shall be able to make Employer Contributions indefinitely, but the Government of Guam shall be under no obligation to continue Employer Contributions or to maintain the Plan for any given length of time. The Legislature of the Government of Guam may, in its sole discretion, completely discontinue Employer Contributions or terminate the Plan by action of the Board of Trustees, and direct the Board of Trustees with respect to the disposition of the Trust.

12.03 Action Required Upon Plan Termination

Upon the termination of the Plan and after payment of all Plan expenses, including any compensation then due the agents of the Board of Trustees and Plan Administrator, all Participants' Accounts shall be revalued according to the procedures provided in Article VI. Accounts shall be distributed as directed by the Plan Administrator in accordance with the provisions of Article VII.

Notwithstanding the requirements of Article VII, if the Plan is terminated, then the Participant's Account shall be distributed to the Participant (or, in the event of the Participant's death, the Participant's Beneficiary) without the consent of the Participant or, if the Participant is married, the Participant's Spouse.

ARTICLE XIII

QUALIFIED MILITARY SERVICE REQUIREMENTS

13.01 General

Notwithstanding any provision of the Plan to the contrary, contributions, benefits, and service credits with respect to "Qualified Military Service" shall be provided in accordance with Code Section 414(u) and regulations thereunder. For this purpose, "Qualified Military Service" shall mean any service in the uniformed services (as defined in the Uniformed Services Employment and Reemployment Rights Act of 1994, Chapter 43 of Title 38, United States Code ("USERRA")) by any individual if such individual is entitled to reemployment rights under such Chapter with respect to such service.

If any contribution is made by the Employer or the Employee to the Plan with respect to an Employee, and such contribution is required by reason of the Employee's rights resulting from Qualified Military Service, then (a) such contribution shall not be subject to any otherwise applicable limitation contained in Code Sections 402(g) (elective deferrals), 402(h) (simplified employee pension), 403(b) (tax-sheltered annuity), 404(a) (deduction limitations), 404(h) (simplified employee pension deduction limitation), 408 (individual retirement accounts), 415 (limitations on contributions and benefits), or 457 (deferred compensation plan) and other limitations as permitted by the Secretary of Treasury, and shall not be taken into account in applying such limitations to other contributions or benefits under the Plan with respect to the year in which the contributions are made; (b) such contribution shall be subject to such limitations with respect to the year to which the contributions relate (in accordance with rules prescribed by the Secretary of Treasury); and (c) the Plan shall not be treated as failing to meet the requirements of Code Sections 401(a)(4) (nondiscrimination in contributions and benefits), 401(a)(26) (minimum participation requirement), 401(k)(3) (actual deferral percentage test), 401(k)(11) (simple plan), 401(k)(12) (actual deferral percentage test safe harbor), 401(m) (actual contribution percentage test), 403(b)(12) (taxnondiscrimination), 408(k)(3) (simplified employee sheltered annuity pension nondiscrimination), 408(k)(6) (simplified employee pension salary reduction), 408(p) (simplified employee pension), 410(b) (nondiscriminatory coverage), 416 (top-heavy plan requirements), or other requirements as allowable by the Secretary of Treasury by reason of the making of (or the right to make) such contributions.

13.02 Employer Contributions

(a) Additional Elective Employer Contributions

In the case where a Participant is entitled to benefits for Qualified Military Service, the Participant shall be allowed to make a Deferral Election ("Additional Deferral Election") during the period which commences on the date of reemployment of the Participant by the Employer and which extends over a period equal to the lesser of (1) the product of three times the period of the applicable Qualified Military Service or (2) five years ("Makeup Period"). The Elective Employer Contributions determined in accordance with such Deferral Election ("Additional Elective Employer Contributions") shall not exceed the maximum amount of the Elective Employer Contributions that the Participant would have been permitted to make under the terms of the Plan during the

period of Qualified Military Service if the Participant had continued to be employed by the Employer during such period and received a designated level of Compensation ("Deemed Compensation"). For this purpose, the Deemed Compensation shall be equal to: (1) the Compensation that the Participant would have received during the Makeup Period if the Participant were not in Qualified Military Service (determined based on the rate of pay the Participant would have received from the Employer but for absence during the period of Qualified Military Service) or (2) if the Plan Administrator determines that the Compensation the Participant would have received during the period of Qualified Military Service is not reasonably certain, the Participant's average Compensation from the Employer during the 12-month period immediately preceding the Qualified Military Service (or, if shorter, the period of employment immediately preceding the Qualified Military Service). The Additional Elective Employer Contributions shall be credited to the Participant's Elective Contribution Account.

(d) Adjustments to Account

The Participant's Account shall not be entitled to the allocation of any earnings with respect to any Additional Elective Employer Contributions before such contributions are actually made to the Plan. Further, the Participant's Account shall not be entitled to the allocation of any forfeiture of accounts under the Plan with respect to the period of Qualified Military Service.

(e) Credit for Service

In the case of the reemployment of a Participant who is entitled to benefits for Qualified Military Service, the Participant shall be treated as not having incurred a Break in Service with the Employer by reason of the Participant's period of Qualified Military Service. Further the period of Qualified Military Service shall be treated under this Plan to constitute Service for purposes of determining the Participant's vested interest in his Account and for purposes of determining the Participant's accrued benefit under the Plan.

(f) <u>Termination from Service</u>

If a Participant who is entitled to benefits for Qualified Military Service is not reemployed within the period required by USERRA, he shall be deemed to have terminated his Service when his leave of absence became effective, or, if later, at the time allowed in accordance with the provisions of USERRA.

13.03 Interpretation

The provisions of this Article XIV shall be interpreted and applied by the Plan Administrator in a manner consistent with the requirements of Code Section 414(u) and underlying Treasury Regulations and Internal Revenue Service pronouncements.

13.04 HEART Act Provisions

(a) <u>Death Benefits</u>

If a Participant dies on or after January 1, 2007, while on a leave of absence to perform Qualified Military Service (as defined in Code Section 414(u)), the Participant's Beneficiary shall be entitled to any additional benefits (other than benefit accruals relating to the period of Qualified Military Service) provided under the Plan as if the Participant had resumed and then terminated employment on account of death, in accordance with Code Section 457(g)(4). In particular, the Participant's benefit under the Plan shall become fully vested under Article V.

(b) <u>Differential Wage Payments</u>

This Section 13.04(b) shall apply for Plan Years beginning after December 31, 2008, if the Employer provides differential wage payments (within the meaning of Code Section 3401(h)(2)) with respect to a period during which the individual receiving such payments is in Qualified Military Service. The individual receiving differential wage payments shall be treated as an Employee of the Employer making the payment. Moreover, the differential wage payments shall be treated as Compensation for purposes of Code Section 415(c)(3) and Treas. Reg. 1.415(c)-2. However, differential wage payments shall not be treated as Compensation for purposes of determining contributions and benefits under the Plan. Thus, no Employer Contributions (whether elective or nonelective contributions) shall be determined or made on behalf of a Participant based on any differential wage payments.

The Plan shall not be treated as failing to meet the requirements of any provision described in Code Section 414(u)(1)(C) by reason of any contribution or benefit which is based on differential wage payments, but only if all Employees of the Employer performing service in the uniformed services, as described in Code Section 3401(h)(2)(A), are entitled to receive differential wage payments on reasonably equivalent terms and, if eligible to participate in a qualified retirement plan maintained by the Employer, to make contributions based on the payments on reasonably equivalent terms (taking into account Code Sections 410(b)(3), (4), and (5)).

(c) Deemed Severance from Employment

This Section 13.04(c) shall apply for Plan Years beginning after December 31, 2008. Notwithstanding the above Section 13.04(b), for purposes of applying rules under Code Section 457(b) that permit distribution of elective contributions after severance from employment, an individual shall be treated as having been severed from employment during any period the individual is performing service in the uniformed services as described in Code Section 3401(h)(2)(A). If an individual elects to receive a distribution of elective contributions by reason of such a deemed severance from employment, the individual may not make an elective deferral during the six-month period beginning on the date of the distribution.

ARTICLE XIV

MISCELLANEOUS

14.01 Participant's Rights

The Plan shall not be deemed to constitute a contract between the Employer and any Participant or to be a consideration or inducement for the employment of any Participant or Employee. Nothing contained in the Plan shall be deemed to give any Participant or Employee the right to be retained in the Service of the Employer or to interfere with the right of the Employer to discharge any Participant or Employee at any time regardless of the effect which such discharge shall have upon him as a Participant under the Plan.

14.02 Construction

The provisions of the Plan shall be construed and enforced according to the laws of the state in which the Employer was incorporated or otherwise established. However, it is the intent of the Employer that the Plan meet the applicable requirements of the Code, and any ambiguities in construction relating to the provisions of the Plan shall be interpreted in order to effectuate such intent.

14.03 Headings

Headings of the Articles and Sections under the Plan are inserted only for convenience of reference and are not to be considered in the construction of the provisions of the Plan.

14.04 Gender and Number

Wherever any words are used under the Plan in the masculine, feminine, or neuter gender, they shall be construed as though there were also used in another gender in all cases where they would so apply, and whenever words are used herein in the singular or plural, they shall be construed as though they were also used in the other form in all cases where they would so apply.

its	IN respective	WITNE duly	OF, the Employer has representatives			s Plan to be 30th	executed by day of	
		June	_·				,	
			BOARD OF	TRUS'	TEES (OF THE		
			GOVERNMI FUND	ENT O	F GUA	AM RETIRI	EMENT	
			FUND	î	i d			
			bes.	S. J. ACV				
		1	JOH T. SAN Chairman	AGUS	TIN	A		
			1,, 2	~	9	_		
			WILFRED P	LEO	N GUE	ERRERO, E	d.D.	
			Vice Chairm	an				
			AShGu	erren	۵			
			ANTOLINA Secretary	S. LE	ON GU	ERRERO		
			Secretary					
			July	1				
			GERARD A.	CRUZ	Z			
			Treasurer					
			Kalg	S				
				'BRIE	EN			
			Trustee		0/	/		
			1.11	14	//			
			ALBERT H. SANTOS					
			Trustee	7				
			Tatal	_				
			KATHERIN	Е Т. Е	. TAIT	ANO		
			Trustee					

A	i i i i i i i i i i i i i i i i i i i